Author's response to reviews

Title: Complicated coeliac disease: two different patterns of natural history, two different prognoses. A multicentre case control study

Authors:

Federico Biagi (f.biagi@smatteo.pv.it)
Alessandra Marchese (ale.mar82@gmail.com)
Francesca Ferretti (francesca.ferretti01@gmail.com)
Rachele Ciccocioppo (r.ciccocioppo@smatteo.pv.it)
Annalisa Schiepatti (salinana@hotmail.it)
Umberto Volta (uvolt@yahoo.com)
Giacomo Caio (gpi.caio@gmail.com)
Carolina Ciacci (cciacci@unisa.it)
Fabiana Zingone (z.fabiana@hotmail.it)
Anna D'Odorico (anna.dodorico@unipd.it)
Antonio Carroccio (acarroccio@hotmail.com)
Giuseppe Ambrosiano (ambrosiano80@yahoo.it)
Pasquale Mansueto (pamansu@libero.it)
Antonio Gasbarrini (agasbarrini@rm.unicatt.it)
Anna Chiara Piscaglia (annachiarapiscaglia@hotmail.com)
Alida Andrealli (alida.andrealli@gmail.com)
Marco Astegiano (marcoastegiano@yahoo.it)
Sergio Segato (sergio.segato@ospedale.varese.it)
Matteo Neri (mneri@unich.it)
Alberto Meggio (Alberto.Meggio@apss.tn.it)
Giovanni de Pretis (Giovanni.dePretis@apss.tn.it)
Italo De Vitis (italodev@tin.it)
Paolo Gobbi (gobbipg@smatteo.pv.it)
Gino R Corazza (gr.corazza@smatteo.pv.it)

Version: 2
Date: 10 March 2014

Author's response to reviews:

10th March 2014

Editor
BMC Gastroenterology

Dear Sirs,

Here enclosed please find an electronic copy of our manuscript “Complicated coeliac disease: two different patterns of natural history, two different prognoses. A multicentre case control study”, that we submit to be reviewed for consideration for publication in BMC Gastroenterology.
The manuscript has not been published previously (except in the form of an abstract or as part of a published lecture or academic thesis), and is not under consideration (in whole or in part) for publication elsewhere.

The authors have no conflict of interests.

Yours Sincerely

Federico Biagi

Dr. F Biagi
Coeliac centre/1st Dept. of Internal Medicine
Fondazione IRCCS Policlinico San Matteo
P.le Golgi, 19
27100 Pavia, Italy
Tel. +39 0382 502973
Fax +39 0382 502618
E-mail f.biagi@smatteo.pv.it